Table 6. Solutions for the mass of (11) Parthenope and the orbital elements of (17) Thetis calculated using diffe-rent values for the mass of (4) Vesta; "s" is the standard deviation of the residuals calculated with each solution;the central value of the mass of Vesta (1.35 10-10 M0) is the value used to calculate the solution given in Table 4Mass ofMass obtainedDifference with the orbital elements of Thetis given in Table 4s (")Vesta(10-10 M0)for Parthenope(10-12 M0)Da(10-9 AU)De(10-8)Di(10-6 �)D W (10-6 �)D w (10-6 �)DM(10-6 �)residuals1.22.5720010-10.851.32.581001000.85 1.352.580.851.42.59-1001000.851.52.60-2001000.85